Author's response to reviews

Title: The burden of varicella from a parent's perspective and its societal impact in The Netherlands: an Internet survey

Authors:

Judith H Wolleswinkel-van den Bosch (wolleswinkel@pallas-healthresearch.com)
Anouk M Speets (speets@pallas-healthresearch.com)
Hans C Rümke (rumke@vaxinostics.com)
Pearl D Gumbs (pearl.d.gumbs@gsk.com)
Sander C Fortanier (sander.c.fortanier@gsk.com)

Version: 3 Date: 24 October 2011

Author's response to reviews:

Dear editor,

As requested I have made the following revisions required for publication:

1. The format of the author list has been changed to 'first name initial last name'.
2. I have added the sentence that all authors have read and approved the final manuscript in the author contributions section.
3. I have removed all track changes.
4. I have rewritten the 'author contributions' into a paragraph.
5. I have rewritten the 'competing interests' into a paragraph.
6. I have made some typographical changes (removed redundant spaces).

Best regards,

Judith Wolleswinkel-van den Bosch